Author's response to reviews

Title: Protective effects of Harungana madagascariensis on the Red Blood Cells

Authors:

Prosper N Biapa Cabral (prbiapa@yahoo.fr)
Horea Matei (hmatei@umfcluj.ro)
Ștefana Bâlici (stefanabalici@yahoo.com)
Julius E Oben (juliusoben@hotmail.com)
Jeanne Y Ngogang (jngogang@yahoo.fr)

Version: 2 Date: 25 July 2012

Author's response to reviews:

Editorial request:

1. Please revise the ethical statements in your manuscript. Please note that you do not have to make a new submission, but simply upload a new version including the revised ethical statements.

Answer
- The scientific committee of the University of Yaounde I and the Cameroon National Ethics Committee approved the experimental procedures.
- The procedure was approved by the scientific committee of the University of Pharmacy and Medicine, Cluj-Napoca, Romania. The subject gave written informed consent for the use of their red blood cells.